Author’s response to reviews

Title: Long-term corticosteroid use and dietary advice: A qualitative analysis of the difficulties encountered by patients

Authors:
muriel nogue (murielnogue@gmail.com)
Jacques Rambaud (jacques.rambaud@wanadoo.fr)
Sylvie Fabre (dr.sylviefabre@gmail.com)
Nathalie Filippi (nat.filippi@gmail.com)
Christian Jorgensen (c-jorgensen@chu-montpellier.fr)
Yves-Marie Pers (ympers2000@yahoo.fr)

Version: 3 Date: 10 Jan 2019

Author’s response to reviews:

Dear Editor, Dear Reviewers,

We are thankful to the reviewers for their in-depth evaluation and relevant comments that will genuinely improve our manuscript. We hope that our point-by-point responses and associated corrections in the manuscript will satisfy their concerns and allow our work to be considered for publication by BMC Health Service Research.

Please note that we included line numbers to facilitate review.

Please let us know if this revised version of the manuscript meets your expectations and if it is acceptable as it stands for the review.

Sincerely,

M.Nogue, YM.Pers.